# A PROPOSED PLAN FOR INTEGRATING HIGH SCHOOL ASSEMBLY PROGRAMS AND SPEECH ACTIVITIES

Thesis for the Degree of M. A.

MICHIGAN STATE UNIVERSITY

G. June Shirley

1959

LIERARY Michigan State University



RETURNING MATERIALS:
Place in book drop to remove this checkout from your record. FINES will be charged if book is returned after the date stamped below.

# A PROPOSED PLAN FOR INTEGRATING HIGH SCHOOL ASSEMBLY PROGRAMS AND SPEECH ACTIVITIES

Бy

G. June Shirley

#### A Thesis

Submitted to the College of Communicati n Arts of Michigan State University
In Partial Fulfillment of the Requirements for the Degree of

Master of Arts

Department of Speech

1959

### ACICKULEXCINERTS

The author wishes to acknowledge her indebtedness and express her sincers thanks to Dr. Fred Alexander, under whose kind supervision and inspiration this study was undertaken.

She is also greatly indebted to Dr. Gordon Thomas for his valuable sesistance and guidance.

# A PROPOSED PLAN FOR INTEGRATING HIGH SCHOOL ASSU BLY PROGRAMS AND SPEECH ACTIVITIES

by

G. June Shirley

Submitted to the College of Communication Arts of Michigan State University in Partial Fulfillment of the Requirements for the Degree of

Master of Arts

Department of Speech
1959

Approved Frederick S. alejandes

#### ADSTRACT

It was the purpose of this study to develop a recommended assembly program which would meet the need of high school administrators. Here specifically, this study was to set up a plan that would successfully integrate the assembly program with the speech activities.

The educational value of the assembly program was established. Research reveals that educators agree that the assembly is a vital force and has an important role to play in public secondary education.

An investigation centered on the assembly practices found in schools today. It was found that some schools have satisfactory practices while many others have serious and disturbing problems. The problems center arounds (1) time for the assembly; (2) piece and equipment for the assembly; (3) planning and organizing the assembly; (4) arranging suitable programs; (5) interest and co-operation of students; (4) public-relations problem; and (7) teacher-training and cooperation. This investigation revealed that even though the assembly program is considered important it is a problem to operate.

The history and development of the assembly program were traced through rather distinct stages. The hypothesis was given that the proposed plan of establishing the assembly

es an academic class is the third natural and logical development in the assembly program.

It was proposed that the assembly become an academic class, for it needs and must have purposeful planning, organising, and producing. This class could be in one of several departments e. g. music, inglish, speech, social studies. It was felt that because of the nature of the speech activities and the speech discipline this class should be placed in the speech department.

Doth as a Dackground and as a guide, educational philosophy, principles of secondary education, and objectives of asscribly programs were established. Using these philosophical assumptions as a guide, cleven criteria were set up for evaluating an assembly program.

Six high school assembly programs were evaluated with the proposed criteria. These schools were chosen because of location, type, and the fact that these programs to some extent attempted to integrate speech activities and assembly programs. Although those programs meet the criteria in verying degrees, none were found to meet the criteria in their entirety.

A program was developed which it is felt will successfully integrate the assembly program and speech activities. The details for operation of this program were outlined.
The program meets ell the criteria. The program has
Limitations and has not been offered as the only solution
to the high school essenbly program problem.

# TABLE OF CONTENTS

| CIAPTI |  | PAGE |
|--------|--|------|
| I.     | INTRODUCTION                                 | 1    |
|        | A. The Purpese                               | 10   |
|        | B. Definition of Terms                       | 11   |
|        | C. Justification                             | 11   |
|        | D. Limitations                               | 12   |
|        | I. The Procedure and Organisation of the     |      |
|        | Thesis                                       | 13   |
|        | F. Survey of the Literature                  | 14   |
| II.    | KESTORY AND DEVILOPMENT OF ASSEMBLY PROGRAMS | 17   |
| ٠      | A. The Frequency of Fregrens                 | 26   |
|        | 3. The Length of Programs                    | 26   |
|        | D. The Time to Held Programs                 | 26   |
|        | E. Planning and Rehearsing Programs          | 27   |
|        | F. The Types of Programs                     | 26   |
|        | G. The Attendance at Programs                | 29   |
| 111.   | EDUCATIONAL PRILOSOPHY FOR ASSEMBLY PROGRAMS | 30   |
|        | A. Principles of Education                   | 35   |
|        | B. Principles of Secondary Education         | 36   |
|        | C. Objectives of Assembly Programs           | 39   |
|        | D. Criteria for Assembly Programs            | þО   |
| IA.    | AN EVALUATION OF ASSESSELY PROGRAMS IN WHICH |      |
|        | SPEECH ACTIVITIES WERE INTEGRATED            | 46   |

| CWPTER   |      |     |       |      |           |      |     |      |             |      |     |   |   | _ |     |   |   | PAGE |
|----------|------|-----|-------|------|-----------|------|-----|------|-------------|------|-----|---|---|---|-----|---|---|------|
|          | A,   | 1   | 2086  | eck  | H         | 1111 | er  | y A  | cad         |      | •   |   | • | • | •   | • | • | 47   |
|          | B.   | ì   | iake  | fie  | 14        | Mig  | jh  | Sch  | 001         | •    | •   | • | • | • | ÷   | • | • | 50   |
|          | C,   | C   | :aum  | ani  | ty        | Mig  | jh  | Sch  | 001         | •    | •   | • | • | • | •   | • | • | 52   |
|          | D.   | I   | lou l | der  | H         | l gh | \$e | hos  | 1           | • •  | ę   | • | • | • | •   | • | • | 54   |
|          | E,   | h   | hi t  | efi  | sh        | Bey  | r N | i gh | Se          | hoo  | 1   | • | • | • | •   | • | ŗ | 57   |
|          | P,   | 1   | hor   | ent. | <b>ed</b> | Mig  | jh  | Seb  | <b></b> 1   | •    | •   | • | • | • | •   | ÷ | • | 60   |
| ٧, ٨     | n ea | M   | LAU.  | 101  | O         | 7 77 | E   | PRO  | <b>P</b> 08 | ED : | Pi, |   | • | • | •   | • | • | 64   |
| VI.      | CHCL | LUS | ia    | •    | •         | P •  | •   | • •  | ♥, ,        | • •  | •   | • | • | • | . 🕈 | • | • | 81   |
| RIM.IOSE | APHY | •   |       | _    |           |      | _   |      |             |      |     |   |   |   | _   |   |   | 86   |

## Chapter I

#### INTRODUCTION

One of the most significant and promising developments in extracurricular activities in high schools is that of the assembly. It offers an ideal opportunity for helping youth to see and understand the problems that must be faced in life. The authors of <u>Planning Student Activities</u> put it this way:

Many high school students are confused by chaotic world conditions, by the changing concepts of the place of religion in their lives, by the new philosophies of disciplines, and by the fact that he needs more money to do the things he wants to do. Adolescents, like adults, need assistance in resolving such problems. \*Potentially, the school assembly is an ideal opportunity for helping youth to make satisfactory adjustments of this nature.

The assembly is the one activity of the entire school program that brings all pupils and faculty together. It is only through school assemblies that the attention of the entire school is focused on common experiences. Fretwell has aptly referred to the assembly as "the town meeting of the school."

This expression implies that the assembly

Franklin A. Miller, James H. Moyer, Robert B. Patrick, Planning Student Activities, New Jersey: Prentice-Hall, Inc., 1956), p. 495.

Elbert K. Fretweil, Extra-Curricular Activities in Secondary Schools, New York: Noughton Millin Company, 1931, p. 208.

is the one appartunity for all to share their experiences, problems, and interests.

School assemblies have been a traditional part of the activities of the secondary school. The evolution of the school assembly has moved through two samewhat distinct stages or pariods.

In the first stage, the anjor cophasis was placed on religious activities. Larly assaublies were called "chapels." Their programs consisted largely of Dible reading, prayer, singing of religious hymns, and often sermons.

As the functions of the secondary schools changed, school assemblies tended to reflect this modification of objectives. The content of the programs changed somewhat. The assembly was now used more to entertain the student body. It was, however, still presided over by the head of the school. Papils participated, for the most part, only as spectators and had little to say about the nature of their assembly programs.

The second stage in the evolution of the assembly which we find emisting in most high schools today is corronly referred to as the "pupil participation stage." In this phase administrators and teachers place increasing emphasis

For a discussion of this and the subsequent stages see: filler, et. al, sp. cit. pp. 475-490.

on active pupil participation in planning and presenting ascending programs. Cloudy but surely assembly programs are becoming student centered in a very real sense.

to the functions of assemblies, educators began to accept
the philosophy popularized by Newey, the philosophy that
"we learn by doing."
They began to look on assembly programs
as an excellent opportunity for developing forms of educational
experiences that had not been fully recognized previously.
Quite naturally the scope of assembly programs became more
comprehensive, so that the assembly today is taking on added
significance in the life of the school. The concept of
its value as a series of learning experiences vastly increases
its potentialities. It produces life-like situations. No
longer are pupils restricted to sitting; they contribute.

with the edvent of this new philosophy came many problems. If the assembly program in our high schools is to realize its potentialities, it must be a well-planned, carefully-organized part of the total life of the school.

<sup>&</sup>lt;sup>6</sup>1513, p. 497.

<sup>5&</sup>lt;u>1510.</u>

<sup>61010,</sup> p. 490.

<sup>7</sup> James W. Whitlock, "Organization and Administration of the School Assembly," School Activities, MWIII (May, 1997), 203.

It is, then, the responsibility of the schools to provide an assembly program of such flexibility and diversity that it meets the needs of the great range of abilities and interests of the pupils. This is a big undertaking.

It is doubtful, however, that the school assembly is fully understood, appreciated, and utilized by teachers, administrators, or students. This is evidenced by the presence of so many problems in connection with the assembly. Hany administrators and teachers do not have the time to plan, organize, and produce worthwhile assembly programs. Students, therefore, do not appreciate the programs because they are not well-planned and well-produced.

Perhaps the greatest obstacle to the development of the assembly program has been the lack of adequate planning. No doubt many schools have solved this problem. Too many high schools, however, neglect the assembly. Their programs are haphazardly planned, inadequately sponsored by teachers who usually are given too little time to prepare a program; they are inadequately produced, and represent little cooperative administrator-teacher-student planning.

Another problem is that of preparing programs which have sufficient variety and appeal to hold the interest of a heterogeneous audience.

Laurence S. Flaum, The Activity High School, (New Yorks Harper and Brothers, 1955), p. 510.

This fact was revealed by a survey made by the authors of The Bulletin of the National Association of Secondary-School Principals. This Bulletin reported the assembly practices and programs of 450 selected secondary schools. In addition to insufficient Variety and appeal, the administrators of the schools surveyed indicated another serious problem was that of arranging time for already everburdened teachers to supervise the preparation of programs. This, interestingly enough, ties in directly with the first problem listed above, that of haphazardly planned programs.

If the assembly is to accomplish its purpose, it must be planned as a vital part of the school's program. It is the duty of the administrator to include the assembly as a vital part of the program of the school. The operation of the plan, carrying out of details, and actually making certain the program is a success is the duty of those charged with this responsibility. Making sure the program is a real part of the school is an ever present problem for administrators.

If assemblies do not become a vital part of the school's program, they become a waste of time for both students and teachers. My own experience in the few years in which I have been associated with secondary schools has demonstrated

C. C. Harvey, et. al., "The Assembly Program in the Secondary School," The Bulletin of the National Association of Secondary School Principals, XXX (November, 1946), 39-43.

that in all too many cases high school assembly programs are not of educational value. Tailing with administrators and teachers in other educational systems, I find that assembly programs in their schools are "poor." This further substantiates the belief that the assembly program must be a vital part of the school program or it becames a "waste of time."

Furthermore, although the assembly is considered of educational value and importance to the secondary school, other research indicated it is being misused in far too many schools, and its operation presents a problem to the administrators. For example, a questionnaire covering "best practices" in occurricular activities made by the authors of the book <u>Flanning Stadent Activities</u> indicates that of the hundreds of schools submitting reports of "best practices" emong cocurricular activities less than two per cent of the schools mentioned school assemblies as worthy of special commondation. This could mean many things, but one plausible assumption might well be that there is much roca for improvement.

It is encouraging to note that at the present time there-seems to be a tremendous interest among administrators and teachers in learning methods which can be used in improving the assembly in order to make this activity a more vital

<sup>10</sup> aller, op. cit., p. 514.

and wholesame influence on the stalents. 11

This interest, in my belief, is bringing about the third stage in the development of the assembly program to make them a part of the regular school carriculum. Although the educational value of assemblies has long been recognized, this movement has progressed slowly. Decoming a part of the school carriculum is the third step. At the present time it is on the border line neither wholly extracurricular nor fully carricular.

Current day philosophy behind assembly programs advocate their use as a laboratory where students may gain confidence in expressing themselves before a large audience. 12 Educators agree, therefore, that the assembly should be integrated with the subject natter of the school. Already there have been attempts to integrate it with various subjects taught in the high school. This new movement to integrate the assembly program into the corriculus stems from the difficulty of planning and circoting the assembly programs. Now, then, can this new phase of integration best be accomplished.

Perhaps the most obvious way to solve the problem of good assembly programs is to integrate it with a subject-matter area. This could be done with several disciplines in the high school. For example, it could be done with the

<sup>11</sup> whithoux, op. cit., p. 286.

<sup>12</sup> Nobert L. Fatton, "Setting All Students into the Act," School Executive, LOUV (November, 1950), 74-75.

music program of the school. Also saws schools have integrated it with the leadership class in the school. Another area where it could be integrated successfully, I believe, is speech. The reasons for exchining associates and subject-matter areas are two-fold. First, it would bring benefits to the school gail its students in thet: (1) it would bring to the school gail its students in thet: (1) it would bring to the students more interesting, better prepared, and better organized assembly programs; (2) it would give more students an opportunity to plan and participate in the assembly programs; (3) it would free the administrator of the direct responsibility of the assembly program; and (1) it would give an opportunity for material from all subject-matter areas to be presented in a good assembly program.

Second, it would bring benefits to a curricular area of the school, in this case, the speech department for:

(1) It would give the speech stadents who are in training to meet the public, practice in getting along with people as they work with others to prepare the programs; (2) It would give a chance for better organization and preparation of speech and drama activities which regularly appear on assembly programs; (3) It would give speech students an opportunity to use their creative talents and ideas in creating and producing programs; and (4) It would afford an excellent motivation and outlet for speech and dramatic class work.

There is need for such an outlet, and it is vital to a successful appears program in a school.

The question may be saided as to may select speech as a vehicle to integrate the assumity with the school carricular.

Any answer to this question necessitates a discussion of the place of speech in education.

Operation appears to be as old as socially organized efforts to instruct the young. It now ranks with the most widely studied subjects in our schools. Its fature rests with speech teachers, ability to demonstrate that it has a substantial contribution to make to the world in which we live.

The report of the Harvard darkittee on deneral Education emphasized that in our secondary schools we need practice in oral as well as written expression throughout the carriculum. 
In order to give practice in oral empression, an increasing number of individual high schools throughout the country are developing integrated approaches. Such of these schools are emperimenting with programs involving skills and knowledge common to more than one discipline. Others are experimenting with continued speech and social studies classes in an effort to give practice in using the speech skills. 

12.

Speech teachers teach the speech skills. These are the parts, the specific objectives of their offert, but the whole

Committee, (Cambridge, 1965), p. 112.

<sup>&</sup>quot;Franklin K. Enower, "Speech for all American Youth," The Bulletin of the National Association of Lecondary School Principals, and I (Junuary, Page), Help.

is more than those elements. This whole, which is the integration of the entire process, is the experience of using these skills wisely in realistic though prearranged group situations under the guidance and critical counsel of a good teacher. The need for providing such experiences is probably the most important objective of systematically organised general educational programs in speech. 15

As a secondary school speech teacher, I have found that it is a problem to find a place to use and practice the speech activities. Speech teachers from other schools with whom I have discussed this problem have the same difficulty. It is my opinion that many schools are faced with the same situation. The fact that schools are experimenting with the integration of speech and other subject-matter areas leads me to believe that neither have they solved the problem of an swilet for the speech activities. There should be an out-iet in the high school for these activities. For as Franklin Knewer has said, "The student should be directed in the practice of speech in varied functional life activities."

In view of this fact, it is the author's belief that these speech activities could be successfully and profitably integrated with the secondary school assembly programs.

<sup>15</sup> Ibid.

<sup>16</sup> Prenklin H. Knower, "A Philosophy of Speech for the Secondary School, "Speech Teacher, I (March, 1952), 85.

In the light of these circulationes, this study is being some: (1) to determine to what degree and how successfully speech activities have been made a part of the assumbly program; and (2) to develop a plan which will truly integrate the two.

# .cfimition of Ter .s

There are four terms which need definition: "proposed plan," "assumbly program," "integrating," and "speech activities."

The first term, "proposed plan," will in this study refer to a method of action or procedure offered for consideration or adoption.

"Assumbly program" is the presentation of an order of exercises, numbers or the like to all the papils and staff of a school or, in some instances, to a considerable proportion of them.

"integrating" is bringing together the parts of into a whole so as to pass from an unstable to a relatively simple and stable condition.

"opeach activities" refers to those activities employed in speech situations to teach the speech skills, e.g. discussion, debate, public speaking, dramatics.

# Justification

Because of the importance of assumbly programs and speech activities in present day educational philosophy, there are a number of justifications for this study. The first of these justifications is that it will provide a plan whereby high school administrators can reorganize their assembly

practices and align them with educational philosophy.

Information on this nature can act as a gaide to the addinistrator in the development of an assumbly program plan which
will give educational, inspirational, and worthwhile assumbly
programs to the stadints and faculty.

The second justifucation is that this study will provide a plan which if used by the high school speech teacher can give an outlet for the speech activities.

The third justification for this study is that with the successful integration of speech with the assumbly programs will come the possibility of integrating other disciplines with the assumbly programs.

The fourth and final justification erises from the distinctiveness of this study. As nearly as can be determined, it is investigating an area and proposing a plan that has not been proposed before by previous theses in the field of speech or education.

# Limitations

This study will be limited to a consideration of the assembly programs of schools in the United States. It will be concerned with secondary schools only. The study will consider no extracurricular activity but the assembly program.

A number of studies have attempted the integration of speech activities with other subject-matter areas in the

school. This stary will be concurred with integrating the speech activities with the assembly program only. The proposed plan will cover one year only and will not work in every high school but can be adapted or modified to fit the situation.

# Prop. Note and Organization of the Thisis

The general procedure to be followed in this thesis will be: (i) Determine an educational philosophy for assumbly programs. The method used to determine this will be to examine books of educational philosophy for general education, ? to compile a list of purposes and aims of assumbly programs as revealed by surveys which were done for this purpose, find from the literature what educators consider to be the purposes of good assembly programs, and by evaluating these lists in the light of educational philosophy for general education compose a criteria for good assembly programs. (2) investigate the possibilities of integrating speech activities with the assembly programs. This will be done by critically evaluating in the light of the proposed criteria any past or present efforts to integrate these activities. (I) bevelop a general program whereby the speech activities can be successfully integrated with the assumbly programs. Ideas for this program will be taken from the investigation of what has or is being done, but the specific program will be original with the eather and the one which the thinks will

best complement both the assuming program and the speech activities.

This thesis has been organized into six chapters as follows: Chapter I presents the introduction to the thesis. Chapter II contains the history and development of the assembly program. Chapter III gives the purposes and aims of good assembly programs according to elecational philosophy. Chapter IV presents an analysis of some present practices of integrating speech with assembly programs. Chapter V contains a proposed plan in accordance with educational philosophy for assembly programs integrated with speech activities. Chapter VI presents a summary and suggestions for further study.

# Sarvey of the Literature

An investigation of speech theses was calle to prevent an overlapping or caplication of research completed or in progress. Sources used in this investigation were "mostoral discertations in Speech born in Progress," "A bstracts of Theses in Speech and Frame," "An Index to Graduate work in the Field of Speech," and justices Theses in Graduate.

<sup>17</sup>J. Jeffery Ader, "moctoral missertations in appeach and Drama," Speech monographs (Annually Since 1951).

<sup>18</sup> June W. Bow, "Abstracts of Theses in Speech and Brama, Speech accorages (Annually Since I) ).

<sup>19</sup> Franchia H. .mower, "An index to Graduate don't in the Field of Speech," (paich : prographs (Annually since 1929).

T. A. Lambe end Herbert H. Silvey, <u>casters Theses in Advantant</u> (Annually since 1951).

The investigation reveals that no study such as this has been or is being done. There are, however, a number of studies which are similar in nature.

George Finley at Marquette University in 1937 discussed the value of using trained suditorium class programs to improve assembly programs. 21 Myrtle Hardy gives the possibilities of using speech arts in suditorium activities of the Platoon School. 22 Perhaps the one most closely related to this study deals with the study of speech activities as a coordinating and educational factor in the secondary school assembly. This thesis, written by G. T. Sellers in 1934, goes into detail to try to prove that speech training and activities if used in an assembly program will aid the education of students. 23

Even farther removed are two other studiess Doxie Hudson has developed a program of activities to be used in

<sup>21</sup> George Finley, "The Value of Trained Auditorium Class Programs as an Aid for Laproved Assembly Programs," (unpublished Faster's thesis, Farquette University, 1937).

<sup>22</sup> Tyrtle Hardy, "The Place of Speech Arts in Auditorium Activities of the Pistoon School," (unpublished Haster's thesis, University of Southern Chilfornia, 1933).

<sup>230.</sup> T. Sellers, \*Speech as an Educational Factor in the Secondary School Assembly, \* (unpublished Faster's thesis, University of Southern California, 1933).

an auditorium class for grades up through seven. 24 Mary
B. King's thesis is concerned with the possibilities of
integrating speech training with the social studies program
of the school. 25

Although these studies deal with the same general area, they do not accomplish the same things which will be the purpose of this study. In addition, it would be pointed out that all of these theses except two were written before 1940.

It is with these facts in mind that the author turns now to the history and development of secondary school assembly programs.

Program, " (unpublished Master's thesis, Southwest Texas State Teachers College, San Marcas, Texas, 1955).

<sup>25</sup> Mary B. King, "A Study Concerned with the Integration of Speech Training with the Social Studies Program to Influence Social Adjustment," (unpublished Master's thesis, Boston University, 1954).

### Chapter II

# MISTORY AND DEVELOPMENT OF ASSLITCHY PROGRAMS

The secondary-school assembly originated from the "opening exercises" or "chapel services" of the early colleges. Decause a chief aim of higher education in the early days was to give training for the ministry, the religious motive predominated in all courses and activities. The daily devotional periods were considered very important.

Even after colleges widened their objectives and curricula, this chapel service was still maintained, not only because it was traditional and was considered a worthy educational activity, but also because it was good advortising to parents of prespective students. It helped to show the parents that the purpose of the college was religious instruction.

The "chapel services" of these early colleges were cerried on in a very formal and dignified manner, with little washation from day to day in the programs presented. The faculty was scated on the stage, and the students marched in to take their assigned scate in a very orderly fashion. the head of the institution was in absolute charge, and the success of the program depended upon his forcefulness in "putting over" his "message" or some "moral truth." In

Herry C. McRown, Assembly and Auditorium Activities, (N.w Yorks MacMillan Coopeny, 1900), p. J.

addition to his exhortations, the program generally consisted of the reading of passages from the Scriptures, prayer, the singing of hymns, and amnouncements.

The scademies, and later the secondary schools, took ever many of the features of the early colleges in which their teachers and administrators had been trained. Naturally, the "chapel service" was one of the activities inherited. It was taken over with little, if any, change or adaptations; entreached itself as part of the program; and tended to perpetuate itself until long after many other espects of the secondary schools had been modernized to meet new conditions and needs.

This "chapel" in the high schools has progressed through very distinct stages. These were referred to briefly in the introduction to this study. A more thorough description of these stages is necessary in order to fully understand the development of the assembly progrem.

there have been smaller innovations which have brought some change within each period. There has been, of course, an everlapping of the periods to a certain extent.

The first period in the development of the assembly in the secondary schools began with the advent of our first schools and lasted until the later part of the nineteenth were of the sermonic and moralizing type similar to the programs of the pioneer colleges. They consisted largely of Bible reading, prayer, singling of religious hymnes, and often sermons. The programs were perfunctory, starsotyped, and monotonous. Attendance was compulsory and discipling was severe. The principal or head master was in complete charge.

About the time of the beginning of the academy movement in the last half of the eightsenth century and about the middle of the first period in the development of the assembly program, the functions of the secondary schools changed. The objectives of Latin Grammar schools had been to prepare a leadership for church and state. The function of the Academy was to teach what people wanted to learn, to train the mind, to provide mental discipline, and to develop character. The school assemblies tended to reflect this modification of objectives. Assembly programs became more comprehensive in terms of program content. The religious objective continued to be of major importance, but in addition, the assembly was looked upon as an opportunity to announce school policy, to recognise superior schievement, to raise the culture level of the pupils, and to entertain the student body. It was during this time that faculty members were delegated partial

William H. French, American Secondary Education, (New Yorks The Coyseey Press, 1777), pp. 50-70.

responsibility for programs. Pupils participated, for the most part, only as speckators and had little to say about the nature of their assembly programs. They were marched into the assembly room and scated in groups that were rigidly supervised by a member of the faculty. Enforcement of discipling was stressed. The general atmosphere all too often suggested that pupils should be seen, not heard.

The "morning exercise" or "chapel" so entrenched itself into the high schools that it was not until after the turn of the century that there was any great change. It was at this time, ushering in the second stage that the name began to change to "assembly". Harry Mexicum in his book published in 1900 points out:

Although the "essembly" is fast supplenting "chapel," as school authorities appreciate its wonderful educational potentialities, many schools will profess to have a chapel service, and not a few still have the faculty scated on the stage!"

The change in the name was significant; for the "assembly" came to mean something more than a more physical assembling of student body and faculty, sawething more than a gathering of minds that were to follow as closely as they could the idea of a speaker. The new news suggested the assembly of

Franklia A. Miller, et. al., Planning Student Activities, New Jersey: Prentice-Hall, Inc., 1900), D. 697.

licion, loc. cit.

the interests and aspirations of the entire student body.5

This second stage in the evolution of the assembly although a gradual process was a major change. It, too, developed as did the change in the later part of the first stage as the result of the new philosophy of secondary schools. Fretzell writing in 1925, says of this periods

in the last decade the assembly in the public high school has changed almost completely in aim, program, character of participants, time and organisation.

As the school has become a social organization, as it has come to be recognised that it is what the learner does, that educates him, the assembly has changed. However, since the stereotype assembly was firmly entreached, required little or no planning or presentation, and since high school principals enjoyed giving advice and had great faith in the efficacy of this advice, this type of assembly tended to perpetuate itself. Other reasons similar to this probably account for the continuance of the old type of assembly long after many other high school practices had fallen in line with the more progressive educational thought.

McKown attributes the slowness with which the first stage gave way to the new to the attitude of the high school principals toward the assembly. He says:

One of the most important reasons for the comparatively tardy capitalisation of the school

School Assembly, " (unpublished Master's thesis, University of Southern California, 1934), p. 1.

E. H. Fretwell, "Extra-Curricular Activities of Secondary Schools," Teachers College Record, XXV (Nay, 1921), 61-69.

assembly as an educational opportunity has been the attitude of the school principal towards it. /Nis attitude is to/consider the assembly merely an administrative device or convenience instead of an educational opportunity.

This is demonstrated by a study made in 1923, of the relationship of the principal to the assembly program.

E. E. Evans stated that out of the ninety-five principals reporting, fifty-three arranged their own assembly programs and about one-half of the remainder were chairmen of their assembly committees, composed usually of teachers, and also that there were few instances of pupil representation. In short, in approximately three-fourths of the cases studied, the principal was the program. He assigned or dictated the program, he scheduled the meeting, and he handled it when it was given, with little or no help from anyone.

Eventually the assembly became an important and integral part of the educational offering of the school. Fretwell praised its educative value.

The need of the assembly as an educative agency is recognized, for pupils form habits and are educated by what they do. Hany a young person is inspired to increased effort by finding he can preside over a meeting, play a solo, interpret a dramatic role, sing a song, play in the orchestra, or that he has something to say and that his companions will listen.

<sup>7</sup>McKown, op. cit. pp. 24-25.

E. E. Evens, "What to Do with the High School Assembly," School Review, XXXI (April, 1923), 282-286.

Association of Secondary School Principals, (Sixth Yearbook, 1922), 147.

This period in the development of the assembly program may be labelled the "student participation stage," a period in which the student organizes, plans, and controls the assembly. Probably a majority of the secondary schools are in this stage or some phase of it at the present time.

Students are invited to cooperate in the selection of program themes and in the preparation of program materials; student assembly chairmen are replacing school officials and teachers; school band and orchestra music, and dramatic activities have a prominent place in the programs; and many programs are staged with little or no direct adult participation. Usually there is an assembly committee consisting of several students and one or more faculty members which has charge of developing an assembly schedule and co-ordinating the programs arranged by various groups.

The full implications of this change were not seen immediately. The mere emencipation of the student did not insure good assembly programs. This certainly was a step forward, but it was many years before assembly programs achieved any degree of distinction in the high school program. Coorgia Finley writing in 1937, thirty-seven years after the se called "beginning" of this period, tays:

In the progressive schools of today there are few assembly periods because the regular school classes . . . [Would be disrupted]. When there is an assembly period, the programs are usually one of three types as follows:

1. drametic

2. advertising a department, as music

The assembly period is used by the principal to explain rules and conduct. The students listen passively, and as a rule this type of assembly program is a waste of time. 10

In the meantime another innovation was that of audience participation. Very definite attempts are made to have the entire audience participate in the program rather than sit and listen to it. The participants from the stage are secondary; to give the audience an educational experience is the important factor. Some of the methods by which this is done are: the audience is asked to indicate by its applause which of two costumes or examples of interior decoration it likes better, or which one of two procedures in courtesy is proper; or to make use of a self-enalysis blank in a program on galdance. The kaynote terms of this phase are "Perform," "Respond," "Choose," "Evaluate," or "Put to practical use,"

This type of audience participation did not become a lasting thing in assembly programs. Today the emphasis seems to be education for both the participants and audience. For

<sup>10</sup> Georgia I. Finley, "The Value of Trained Auditorium Class Programs as an Aid for Improved Assembly Programs," (unpublished Master's thesis, Marquette University, 1937, p. 106.

<sup>11</sup>C. C. Harvey, et. al., "The Assembly Program in the Secondary School," The Bulletin of the National Association of Secondary School Principals, NOX (November, 1945), h.

<sup>12</sup> McKonn, op. cit. p. 5

today we have a combination of student participation and audience participation. The audience participation is somewhat different from the type mentioned above. For example, we have at pep assemblies students performing and the audience participating when a school yell is given, the school song is sung, or the flag salute is given.

The assembly today is catching up with the trends in the development of other aspects of secondary education. It contributes to the growth of the individual in a democratic environment. This is not to say that all problems in assembly programs have been solved, far from it.

Many problems do exist; thus the reason for this study. In order better to understand and criticise the present day essenbly program, we must examine some of the practices in regard to it.

It is important to know something about the (1) frequency of the assembly programs; (2) length of the assembly programs; (3) time of holding assembly programs; (4) planning and rehearsing the programs; (5) type of programs; and (6) attendance at programs.

that are the practices of the schools? An ensuer to this can be found in two surveys made in  $1945^{13}$  and  $1957^{14}$ 

<sup>13</sup> Harvey, op. cit. pp. 5-6.

Ilijames W. Whitlock, "Organization and Administration of the School Assembly," School Activities, XXVIII (Nay, 1957), 283-286.

although greater reliance will be placed upon the later of these surveys because of its recency.

Frequency of program: The information gathered in this survey indicates that the plans of many schools are some-what flexible. The most common arrangement is the weekly assembly. The daily assembly exists in only three schools and assemblies are held less frequently than one a month in only thirteen of the 389 schools surveyed. The larger the school the greater seems the tendency toward holding the assembly once a week.

Length of the assembly: The length of the assembly periods are varied. Some schools vary the length of the assembly to fit the occasion. In two hundred five schools assembly period is at least forty-five minutes or the periods range from forty-five to sixty minutes in length. Only eighteen schools have assembly periods of less than thirty minutes. Fifty per cent of the school officials responding to this survey feel a need for an assembly period approximating in length the regular class period.

Time to hold assembly: Friday and Wednesday are the days most preferred for holding the assembly. Friday's selection might possibly be due to the fact that most athletic contests are held on that day and would be the best time in the week to held "pep" meetings.

The period in the day for holding the assembly was:

first choice, the last period in the school day; second choice, the first period in the school day; third choice, the third period in the school day. Heny schools have a rotating or floating period, replacing the first period for the first time, the second period, the second time, etc. with each period in succession.

Finning and rehearding the programs: The survey indicates that the arrangements of many schools for planning and rehearsing the programs are not too natisfactory. In one hundred three of the schools responding, programs are planned and rehearsed only outside school hours. Thenty-one per cent of the schools use only the activity period for this purpose. A few use both the activity period and after school hours. It was mentioned in the answers from a large number of schools that the problem of arranging time for the rehearsal of programs is one of the most difficult sace.

The majority of schools in this study have an essembly calendar which is planned in advance.

In the 1745 survey 15 of the schools studied two hundred thirty-four had an assembly committee which had charge of arranging the assembly schedule and coordinating the programs. In several schools, an organization existed known as the

<sup>15</sup> Harvey, op. clt. p. 15.

"Assembly Club" or the "Auditorium Club." One high school had an assembly training class in the speech department which carried on the duties usually performed by the assembly committee.

In many schools today, the student council serves as the essembly complete. This committee is usually composed by both students and teachers. None schools have a director of student schivities who is in charge of assemblies. Some lighten the load of one teacher so that she may have time to take charge of the assembly programs.

The of programs. The trend is every from the "paid" assembly programs. These are usually programs secured from agencies and too many times they are not satisfactory. The author's experience with paid assemblies has been that too many times they are below par educationally and many times are not even entertaining or inspiring. Today most students can get better entertainment on television. In many schools students originate the ideas for the programs. Some schools exchange assembly programs with neighboring schools. These

Catalde speakers, especially from the community, are used. Commercial groups (such as Bell Telephone or General Electric) sometimes present programs. Schools usually present

programs for special days, events, or weeks. These consist of music, drama, speech damanstrations, or a combination of these.

Attendance at programs in most schools attandance at assembly programs is compulsory for both students and teachers. Teachers are usually assigned disciplinary duty during the program. Sometimes students are assigned certain seats and attendance is checked by home room teachers or class sponsors. Faculty supervision is decad necessary at the assembly in most schools.

Cur schools have traveled a long road since the entire student body marched to the suditorium, on signal and with plane accompanished, to hear a "serion" by the principal.

There is still a long road ahead to realize all the potentialities that his within the accombly program.

In the majority of the schools today the essembly is used mainly for extracurricular purposes, but much progress has been made in integrating it with curricular activities. When this is done then the assembly program will have achieved full status in the secondary schools.

#### Chapter III

#### SOUCATIONAL PHILOSOPHY FOR ASSOCIATY PROTERTS

In order to criticize the present system for exercing the essenbly program, we must first exemine its underlying philosophy. What are the principles by which high school assembly programs operate?

If we know the philosophy of assembly programs, we can then apply its principles to see if the present system is accomplishing its purposes.

To develop primary educational philosophy for examply programs and develop a criteria by thich they can be evaluated is the purpose of this chapter. In order to do this it is necessary to determine the importance of a philosophy and the underlying philosophy or principles of education and in this framework determine the purposes of assembly programs.

Fhilosophers have always been concerned not only with attempts to understand the world as it is, but also with discovering how life should be lived, what non should strive for, and how he can improve his social organization.

In every phase of life - material, spiritual, lay, professional - we believe certain things about the activities we perform. These beliefs, usually to a far greater extent than we realize, not only reflect our day-to-day activities but in turn mold and direct these activities.

There is nothing more necessary to mental hosith, to personal and social harmony, to consistency of action and purpose than to be fairly well satisfied with the beliefs that underlie every day conduct. They constitute a type of pattern that enables us to proceed with some degree of efficiency, orderliness and confidence in what we are doing.

Philosophy is a systematic body of principles, after with implications for their practical application to the affairs of life. It is a set of values. Brancid says that philosophy is "the supreme instrument man has fashioned by which, through the ages, he comes to terms with himself as he struggles to organize his existence within the culture."

To him philosophy is the expression of beliefs. To Doughton, philosophy is "not regarded as the summation of knowledge, but as the correlation of knowledge." He says that the function of philosophy is to affect a coherent synthesis of knowledge, to achieve a believe of knowledge, not a totality of wisdom.

Philosophy is the correlation by each man of all his knowledge and experience which may have a bearing upon any problem of life that he may be trying to solve.

Theodore Brameld, Philosophies of Education in Cultural Perspective, (New Yorks Dryden Press, 1955), p. 21-

Abido po 40

Packground (New Yorks Appleton-Contary-Crotts, 1935).

he may say, then, that philosophy is the fusion of one's knowledge and experience that causes one to act the way he does, to hold the standards, and values that he holds, and to have the expirations that influence him. Everyone has some kind of philosophy that preconditions him for certain ways of thinking and behaving.

Philosophy can be used in many mayor inefficiently or efficiently, vaguely or clearly, hapharardly or systematically. We ail philosophize thenever we try to express the things we believe about our lives and about our relations to the rest of life. Educators are certainly no different from others in holding beliefs and utilizing them.

Educational beliefs are the unavoluble concern of everyone connected with the schools. Since element everyone is concerned with the schools we have an immunes empurit of philosophising expressed in the schools.

it is, therefore, necessary to develop a philosophy of education and to set down the guiding principles to give us direction in educating the people of our country.

It is essential in a democracy that both leaters and those led have a clear philosophy of life and a clear philosophy of education.

William H. French, American Secondary Education, (New Yorks The Odysacy Press, 1957), pp. 475-470.

Spaceld, on cite, on Ele

William M. Kilpstrick, Philosophy of Education, (New Yorks Profillian Company, 1951), p. 6.

A philosophy of education is a specialized and limited philosophy. In America it must be a philosophy of democracy conceived as a way of life. It conserns itself with when we think and do about the education of people, and the reasons we think and do these things. It must, of course, be compatible with and consistently constructed within the framework of the larger philosophy which one has adopted.

This philosophy of education calls for organizing principles for education which make it a discipline of character for exabership in a democratic society.

The principles must be practical principles, by which practical policies and particular practices can be directed and Judged, for educational purposes are always relative to the ends of living nor can they be understood apart from life itself, for aducational also grow out of life's alms. 10 Without these definite also esucation would be chaotic rather than underly. It would be impossible to formulate any galiding principles which would give direction to all activities.

Tephraim V. Savers and Mard Madden. Education and the Democratic Feith: An Introduction to Philosophy of Education, (New Yorks Appleton-Century-Croits, 1959), p. 3.

French, op. clt., p. 496.

Severs, 02. cit., p. 14.

<sup>10</sup> Stella Van Petten Henderson, Introduction to Philosophy of Education, (Chicagos University of Chicago Free, 1977), p. 12.

Radyard R. Bent and Heary H. Kronenberg, Principles of Secondary Education, (New York: McGras-Hill, 1955), p. 31.

A philosophy of education is meaningless, however, if it is detached from the circumstances of time and place, because education is a practical enterprise, closely connected with the changing conditions of community life. 12

Bossing has conceived the general function of education

... the adjustment of man to his environment, which contemplates men's adaption to and the reconstruction of his environment to the end that the most enduring satisfactions may accrue to the individual and society.

This is the basic general philosophy of American education.

Henderson sees the task of education a little more specifically as that of guiding

. . . the growth and development of boys and girls so that they will become competent and social-minded citizens of our commonwealth; so that they will work for the commonwealth, not just for their own narrow ends; and so that they will be a more intelligent, and altruistic generation than previous ones.

Our general philosophy of education is, then, that education should promote our common welfare through helping boys and girls to grow in capacity for intelligent and social-minded participation in group life, by helping each child develop into the finest human being he is capable of becoming, able to live life to its fullest and to do his part in making

<sup>12</sup> Mortimer J. Adler and Milton Mayer, The Revolution in Education, (Chicago: The University of Chicago Press, 1950), p. 57.

<sup>13</sup> Melson L. Bossing, Principles of Secondary Education, (New Yorks Prentice-Hall, Inc., 1949), p. 204.

<sup>1</sup>h denderson, loc. cit.

that possible for every other human being. 15

French sees American education as

. . . education which seeks to meet the common needs of youth for competence as a person and as a citizen. 16

These statements of our present day philosophy of education all reflect the same point of view, but they are all quite general. In order to make our philosophy practical, we must set up some principles by which we can put this philosophy into practice.

The White House Conference on Education meeting in Washington, D. C. in 1955 listed the principles of education to be:

- To develop the fundamental skills of communication - reading, writing, spelling, as well as other elements of effective oral and written expression; the arithmetical and mathematical skills, including problemsolving.
- To develop appreciation for our democratic heritage.
- To develop civic rights and responsibilities and knowledge of American institutions.
- To develop respect and appreciation for human value and for the beliefs of others.
- To develop the ability to think and evaluate constructively and creatively.
- 6. To develop effective work habits and self-
- To develop social competency as a contributing member of his family and community.

<sup>15</sup> Henderson, op. cit., p. 238.

<sup>16</sup>will French, et. al., Behavioral Goals of General Education in High School, (New York: Russell Sage Foundation, 1957), p. 39.

8. To develop ethical behavior based on a sense of moral and spiritual values.

 To develop intellectual curiosity and eagerness for life long learning.

- To develop appreciation and self-expression in the arts.
- 11. To develop wise use of time, including constructive leisure pursuits.

12. To develop physical and mental health.

13. To develop understanding of the physical world and men's relation to it as represented through basic knowledge of the sciences.

the To develop an awareness of our relationships with the world community.17

present day philosophy of education, we have developed principles specifically for secondary education. Since this study is concerned with the secondary school assembly, and its function in the secondary school these principles are important to the development of the principles of secondary school assembly programs.

Hand, Professor of Education at the University of Illinois, has outlined nine principles which underlie public secondary education in America:

1. The American public secondary-school is obligated to promote the general welfare, to assist in teaching youths whatever they need to learn during adolescense in order to carry on those basic social processes on which the strength of the United States and the well-being of each citizen depend.

 To promote the general welfare most fully, the public secondary school must build on the past

but orient itself to the future.

<sup>17</sup>will French, et. al., Great Issues in Education, (Chicagos The Great Books Foundation, 1956), pp. 20-21.

3. If societal strength and individual well-being are best to be ensured, one of the top-level priorities of the public secondary school must be to teach youths to think and to communicate.

is societal strength and individual wellbeing are best to be ensured, one of the top-level priorities of the public secondary school must be to teach youths to understand, appreciate, and live the democratic way of

life.

if societal strength and individual well-being are best to be ensured, one of the top-level priorities of the public secondary school must be to teach youths the principles of safe and healthful living, help them develop bodily vigor, and provide a safe and healthful

physical environment.

tasks most effectively, the public secondary school must have youths make a direct study of the problems of making a living, strengthening the family, and providing for physical security, spiritual growth, aesthetics, recreation, and social organisation and control.

 To promote the general welfare most fully, the public secondary school must serve equally well all the educable youths of the community.

 To premote the general welfare most fully, the public secondary school must make full use of what is known about the learning process.

9. To make it a public institution in the best sense, the public secondary school must have its basic education policies shaped by the lay public with competent professional advice, its technical operation determined and interpreted to the public by competent professional efforts.18

These principles are guides to action. Principles are useful only to those who know how to apply them, how to

Herold C. Hend, <u>Principles of Public Secondary Education</u>, (New York: Harcourt, Brace Company, 1958), pp. 74-78.

analyze individual circumstances in which the principles abould be made operative.

Within this framework of educational philosophy and the specific principles of secondary education, educators must place the assembly program. What then, are the principles and purposes of assembly programs in the secondary school?

lescarch reveals that a number of surveys have been made in which secondary school principals listed the primary aims or purposes of assembly programs. There schools responding to the surveys summarized their assembly aims as being identical with those of secondary education.

Educators have formulated what they believe the important objectives of assembly programs should be. These objectives are in essence the same.

With this fact in mind the author has formulated a composite statement of purposes for the gosenbly, based

<sup>19</sup> See: C. C. Harvey, "The Assembly Program in the Secondary School," The Pulletin of the National Assemblies of Secondary School Principals, Act (Assemble, 1,45), 7-11. Las James W. Whitlock, "Purposes of High School Assemblies," School Activities, X Will (April, 1,57), 247-253.

See: Harry Halloum, Assembly and Ambithrium Activities, (New Yorks The Laciditan Calpaly, 1950), pp. 5-11. and Franklia Lilier, et. al., Planning Statent Activities, (New Jerseys Prentice-Hall, Inc., 1950), pp. 470-505. and Louis R. Hiller, et. al., Allied Activities in the Secondary School, (New Yorks Harper and Louisies, 1955), p. 125.

upon actual practices, voiced ales, and desirable educational outcomes. The objectives of actual practices were taken from the proviously exentioned surveys. The voiced nice are a compilation of objectives thought to be important by educators mentioned carlier. All these have been viewed in the light of the preceding educational philosophy. These objectives are: (1) To provide for pupil participation in creating and organizing programs, (2) To help popils to gain ease, poise, and ability to speak effectively before a group of people; (3) To efford description experiences that will raise individual and proup standards; (i) To carich the educational and cultural experiences so that socio-economic and spiritual horizons will be broadened; (5) To develop desirable mudience habits; (6) To sid in unifying the school; (7) To supplement curricular activities and consolidate the entire program of the school; (8) To foster understanding and lovel relationships in the community.

With these objectives, the assumbly is a fundamental and indisputable elecational force. With basic and far-reaching alos, the assumbly can become a vital and influential experience individually and socially.

As previously stated the secondary purpose of this chapter is to develop criteria by which assembly programs can be evaluated. To establish standards of a good assembly

program, an extensive study was made of descriptions of bundreds of assemblies throughout the United States in all sizes and types of schools. The judgment of high school principals as revealed in survey questionnaires was drawn upon. Correspondence and consultation with directors of stadent activities and editors of professional association publications were weighed. The faculty of a large city school, each of which had had experience with assemblies, criticized the criteria formulated. The criteria were then scrutinized carefully to eliminate any points in conflict with a widely used curriculum evaluating instrument. I Finally, those characteristics which seemed important enough for schools to report and unluable enough for educators to stress were simplified to make them more readily usable.

It was from this information and these criteria the author formulated the following criteria which she feels are the most significant to this study:

# Criterion Cne

A good essembly has a regularly scheduled time for planning, releasesing and producing.

Checklist for the Ton Imperative Needs of Youth, Intional Association of Secondary Principals, based on Education for All American Youth, Educational Folicies Commission, National Education Association, Mashington 6, D. C.

Rellie Zetta Thompson, Vitalized Assemblies, 200 Programs for all Cocacions, (New York: L. r. Dullum Company, 1952), pp. 13/-101.

#### Criterian Two

A good assembly has adequate equipment and facilities.

## Griterion Three

A good assembly is suitable for the length of time, the place, the facilities, the purpose, and the occasion.

#### Oritorian Four

A good assumbly is planned and rehearsed with a definite purpose in mind by stadents who are supervised by a well qualified and cooperative teacher.

## Criterion Five

A good assubly is well organized and well produced thus reducing disciplinary action to a minimum.

#### Criterion Six

A good assembly places emphasis on creativity, resourcefulness, and variety by utilizing all the activities of the achool.

## Criterian Seven

A good assembly emphasizes student participation and control.

## Oritorion : ight

A good assembly is interesting and answers a definite need of the students.

# Critorion Mine

A good assembly enriches the educational experience of the participants and addience and stimulates their imagination.

# Criterion Ten

A good assembly engenders wholesome and nutually beneficial community-school relationships.

These criteria cover almost all the aspects of assembly programs. The author feels, however, that an eleventh

criterion is necessary to insure that the first ten will be measured up to.

Most of the problems mentioned in the 1746 survey of assembly practices add up to the lack of direction and organization for assembly programs. 23

In the above mentioned survey predictions for the future of assembly programs were listed by the participating high schools. These predictions reflect the desires of administrators and teachers. Significant predictions were: (1) The assembly will come to occupy a more important place in the life of the school, (2) The assembly will be better planned with more use being made of student ideas, initiative, talent, and energy. (3) the assembly will be regarded as a part of the regular school-wide programs not extracurricular. Statements such as these were made: "Assemblies will be used more to teach." "They will become more and more a part of the curriculum, presented regularly, and tied in with and an outgrowth of class work." "[They will be] incorporated into general or regular school work." (4) Better integration of the assembly with the remainder of the school. There seems to be a general feeling that the assembly is not integrated as well as it should be with the remainder of the school. particularly with classroom activities. One such statement

Ellarvey, co. cit., p. 1.3.

Marvey, op. cit., p. 52.

was: "[The assembly] will be more closely tied to classwork; used more as a laboratory for speech arts, vocal and instrumental music and other areas."

These predictions reflect the desires of administrators and teachers and point also to a need for direction and organization of assemblies. This need demands the eleventh criterion.

#### Criterion Eleven

A good assembly must be the outgrowth of a regularly constituted academic class, the members of which plan and operate the assembly plan.

 The members of this class, supervised by a qualified teacher, plan, organize, and manage the program in addition to training those who will participate in the programs.

 The programs, aithough supervised by an academic class, are all-school assemblies with the participants and program material coming from all areas of the school.

The author believes there are many reasons why the assembly planning must be the specific responsibility of an academic class.

First of all, it gives direction and organization to the assembly program. The assembly program no longer must be shifted from place to place for preparation and planning. Criterion eleven gives stability, importance, and dignity to the assembly program and establishes it as a part of the curriculum.

In addition, the writers of The Bulletin of the National Secondary-School Principals make this statement about its

being a part of the curriculum: "Schools which have tried to integrate the assembly with the curriculum have found that it enriches both the assembly programs presented and the classwork with which they integrate."

The assembly must be a class included in the curriculum, for this is what administrators and teachers want and forses as a way to give to the school better assembly programs. The survey mentioned above gives those statements by administrators and teachers of what they desire and forses for the assembly: "Progressive schools will have one teacher in charge of all assemblies. This teacher will devote all her time to assembly programs and oral English." "[There will be an] auditorium teacher with an auditorium period for each group so that each may have an opportunity to be on a program."

Planning assembly programs through an academic class will motivate students to do better class work, to present better assembly programs, and to regard the assembly as both interesting and worthwhile.

In the author's opinion, the eleventh criterion brings about the third major change in the development of the high school assembly program. She feels that it is inevitable.

Although it is not a criterion, the author believes that the proper place for this academic assembly class is in the speech department. It has been pointed out earlier in

<sup>25</sup> Herrey, op. cit., pp. 4-5.

this paper that this class could be a part of other departments of the school. There are reasons why the author feels that it should be in the speech department. These reasons have been listed in the introduction to this paper.

With this philosophy of present day secondary school essembly programs in mind and using the proposed criteria, we can evaluate the present assembly practices.

#### Chapter IV

# AN EVALUATION OF ASSURELY PROGRAMS IN LINEAR EXPERIM ACTIVITIES NEED INTEGRATED

Since criteria for evaluating assembly programs have been set up, the next task is to see if existing assembly programs meet this criteria. Extensive research (suggests) that many programs do not meet the criteria that have been set up. The purpose of this chapter, therefore, is to point out in what ways these programs do or do not assert these criteria.

Christian there are far too many assembly programs to be discussed here. The programs from six schools have been chosen. They were chosen because the author found after surveying the literature that each of these programs apparently involve integrating to some extent speech activities with the assembly programs.

The author realises that she could not possibly have examined all the programs, and that there may be other programs which would have been better to use for this evaluation. Various programs have been eliminated for definite reasons. The auditorium class program has been emitted, for the author feels that this is an entirely different type of program. The auditorium class was a one semester class all students were required to take. Programs

were prepared and presented by the students. The audience was the remaining class members who had no part on that particular program. All were given an opportunity to appear on a program. It had no connection with the school's assembly program. Also, nowhere in the research was it found that this type of program exists today.

Programs from the following schools will be used: (1)
Peaceck Military Academy, San Antonio, Texas; (2) Wakefield
High School, Arlington, Virginia; (3) Community High School,
Granite City, Illinois; (4) Boulder High School, Boulder,
Calerado; (5) Whitefish Bay High School, Milwankee, Wisconsin;
and (6) Shorewood High School, Shorewood, Wisconsin. The
author feels that this is a cross section of schools from the
United States. Also they are representative of different
types and sizes of schools.

Each school's program will be evaluated separately in the light of the proposed criteria. It should be pointed out that the programs evaluated here are perhaps the better programs from the schools. Also the information available for these programs is by no means complete. It is, therefore, impossible to determine if these programs must some of the criteria.

Peacock Military Academy has the "Open Forum" type of seeembly each Friday of every week. The programs grow out

of classroom situations. All programs are built around current problems and issues of world-wide significance. These problems and issues are discussed in an understandable way by the students. Plans for the program originate in the principal's office. Each member of the faculty assumes the directorship of the weekly programs at a stipulated time. The director decides on the number of topics to be used. Tapics to be discussed are assigned to the necessary number of students by the first-period home-room teacher upon his receipt of the selected topics from the director. The essignment to the student is made on the rotation system, so that by the end of the year the majority or all students have had the opportunity to speak from the stage. Two students appeinted by the director, give a brief recepitulation at the conclusion of the discussion. A student is designated to act as chairmen.

In conjunction with this program a short movie is sometimes used. This movie parallels the content of the program. At the conclusion of the recapitulation, the chairman opens the meeting to the student body. Opinions may be voiced and questions may be asked of the speakers.

The students are encouraged to relate their talks from memory, and they may or may not use the public address system. There are no announcements on the program. The program consists of: a song, pledge to the flag, film, speeches, recapitulation, and forum.

It is obvious that this type of assembly program meets some of the criteria. This is a regularly scheduled program, planned, and produced at a certain time. There is no mention of rehearsal time. If there is rehearsing, it is probably done during the school day. This program is somewhat interesting and meets the needs of some of the students. It does use student participants. It seems from the available information that it is well organised and well produced. Hention was made that discipline was minor. It, also, meets criterion nine to a degree in that it is educational and probably stimulates the imagination of some of the students.

The program, however, does not meet some of the very important criteria. Criterian four is very definitely violated. This program is not planned by the students with a definite purpose in mind. The program, it seems, is planned entirely by the staff and the parts are assigned to the students. Criterian six is obviously not met. Since the programs are the same type every friday, are planned by the faculty, and utilize mostly speech activities, they do not emphasize creativity, resourcefulness, and variety. It does not appear that the programs produce beneficial community school relationships.

To the author these are serious lacks and could prove to be fatel to the operation of a successful assembly program. A student-staff assembly committee at Unhefield High School in Arlington, Virginia, meets early in the year to plan the year's programs. The committee is composed of a student representative, who is a member of the Student Council, from each grade level, the teacher chairman of each grade level, and the administrator in charge of student activities.

This committee attempts, within the time restrictions imposed, to plan programs which serve the purposes outlined. Once the year's programs are set the mechanics of conduct becomes the responsibility of the administrator and his secretary. They handle the details. Formerly, this school used a teacher as coordinator of assemblies. She was given released time from one class. This did not work for two resease. It cost too much money, and it was too much work for the teacher to do in the allotted time.

The students have assigned scats for the progress. They are given, by the Student Council, a guide for proper behavior at assemblies.

The etudents present and participate in the programs.

They are given, by the Student Council, a guide for proper behavior at assemblies.

The students present and participate in the programs.

The types of programs are: sports assomblies, student
powermount, special days i. c. Thenkagiving, Christmas, and

Easter, science demonstrations, and special birthdays i. e. Washington and Lincoln.  $^2$ 

From the available information, it is difficult to determine if this program meets many of the criteria, such ass one, two, three, five, and ten-

programs are definitely planned by the assembly committee composed of students and faculty. This meets criterion four to a certain extent. Even though the program is partially planned by students and supervised by faculty members, we do not know if these faculty members are really qualified for this job. A well qualified supervisor is quite important.

The program seems to utilize most of the activities of the school, the students participate, broadly speaking the program enriches the educational experience of the students, and the programs appear to be interesting.

This program, however, has some very definite famile.

Criteria four is not measured up to fully in that even though initial planning is done by students the detailed planning and supervising is done by the administrator and his secretary. This loss, the author believes, takes much from the educational experience of the students. No mention is made of rehearest time or the purpose of assembly programs.

Marold M. Wilson, "Programs with a Legraing Function," School Executive, 76(November, 1956), 71-72.

Chricusty this school, as do many schools, has trouble finding time to plan these programs. The committee attempts to plan them "within the time restrictions imposed." They tried releasing a teacher one hour per day to do this and found that it was too much work for her. This is an inedication to the author that there is no definite time for planning and rehearsing the program, and that the creative ability and resourcefulness of the students are not fully attitized. There is no mention of who is responsible for individual programs and whether each one is well organised and well produced. These shortcomings can make the difference between a successful and unsuccessful assembly program.

At Community High School, Granite City, Illinois, the essembly committee is composed of faculty and students with the speech teacher as coordinator.

The history class has decided to give a program. About two weeks prior to the scheduled time for the program, the speech teacher meets with the teacher of the history class and a class committee. Various ideas are presented and talent in the class is considered. The speech teacher guides the student committee in the formation of an attractive program. The program is cast from the members of the class. The program goes into rehearest and as many members of the class participate as possible. The title of the program was the Mold These Truths." An original script was prepared by

the nerrator; a group presented an original play; a boy read a cutting from "The Patriota" by Signay Kingsley; a girl gave an original speech; and a voice from the past read "The Presmble of the Constitution" over the public address system.

in the author's opinion Community High School comes a little closer to solving the problem of the assembly program than the previously mentioned programs in that the chairman of their assembly committee is, all things being equal, a well qualified person, the speech teacher.

This practice partially fulfills criterion four. The programs are rehearsed during the school day by the teacher of the class giving the program. It would seem from the program listed above that they are planned with a definite purpose in mind and student participation is emphasized.

The program seems to be well organized, but it is not known if the program is well produced. Whether the speech teacher trains the participants for the production is not known. Would the speech teacher have time to do this? This emission could down the program to failure.

The creativity and resourcefulness of the students are partially utilised in that the students in the class or club

Fiery Blackburn, \*The Speech Teacher and the High School Assembly Program, \*The Genterly Journal of Speech, 33(February, 1927), 60-62.

giving the program help write and propage the programs. The author feels, however, that criterion six has not been met fully. The job of the speech teacher of helping to create the ideas for these programs could and should be done by students trained and supervised by the speech teacher for this task. This would more fully utilize the abilities of the students. Information as to whether the programs have variety and utilize all the activities of the achool is not available. The program listed above appears to be somewhat monotonous in that there is so much reading. The author feels that the idea is good, but that it is not as interesting as it should be. The program has some educational value, but it is doubtful whether it would stimulate the imagination of the students. It should be pointed out the variety by using all the school's activities and an interesting and stimulating program is important.

The assembly committee at Boulder High School, Boulder, Colorado, achedules the assemblies and chooses the purposes of these assemblies. This committee is composed of faculty numbers and Student Council numbers who represent all high achoel classes.

The committee reviews last year's programs. They then present a ballot to the students and ask them to evaluate the programs. The types of programs are chosen from the

results of the balloting. The schedule is them set up. The schedule contains the date of the program, the nature of the program, and the sponsoring group. Conscionally on the schedule there is an open date. The coumittee sees, however, that it is filled two weeks in advance. The committee schedules the programs so there is variety and that similar types of programs are spaced so as not to fail on succeeding tweks. Representatives from the assembly coumittee preview and attend the rehearsals for assemblies.

ilers is a typical program. This program was built around the television-radio idea, the program supposedly coming from Station "DUS," The acts consisted of the girls of the triple tris appearing in Mestern Costanes singing comboy songs around a campling, four couples from the Spanish Club dencing tango numbers, a three man tumbling team, a magician, the return of the triple trip - this time in evening dresses - singing popular and semiclassical songs, a comedy skit, and a boy who did a show-stopping (and ending) boogle-woogle number on a bot trumpet.

in the program at Boulder High, the emergence of an important factor is seen. This is the practice of

Theodore Skinner, "Suggestions for the High School Assembly," The Cuarterly Journal of Speech, SS(Lecember, 1747), 515-520.

representatives from the assembly committee previouing and attending rehearsals for assembly programs. The author feels that this is getting close to the basic problem of assembly programs. It is not known, however, whether these representatives are trained in judging good programs or if they have anything to say at the rehearsal that would improve the program. It is important that these representatives be trained in program building and know how to help others to perform well on a program.

This program meets part of criterion four in that the programs are planned by students. We do not know if the family supervisor is a well qualified person or not. No mention is made of when the program is reheared, but we do know that the committee plans each program with a purpose. However, since we don't know who sees to the actual developing, organizing, and producing of individual programs, we cannot be sure these purposes have been carried out.

The Boulder High program does emphasize student participation. The assembly committee tries to work variety into the programs. After examining the program listed above, however, the author believes that it perhaps was interesting and stimulating, but its educational value is doubtful.

Decembe of insufficient information, it is not known whether the creativity and resourcefulness of the students

are used. The author feels, however, that Doulder High School has made a start toward producing good assembly programs, but that they stop short of the goal.

The Student Council committee, advised by one of the speech teachers, each spring plans the assemblies for the fellowing year at Whitefish Bay Migh School, Mituaukee, Wisconsin, Claseroom teachers are contacted for their approval as to dates and participation. Some outside acts are signed.

Each section of sophomore speech is represented in an assembly program and is allotted a certain amount of time. Each section has its own chairman. As far as possible the selection of the acts to be presented is a matter for class decision. This stimulates discussion of sudience interests and the smitsbility of materials for school assemblies. Students become source of the need for selecting acts performed by speakers who have voices which will project. Poorly written scripts are ruled out. Each speech class seeks to outdo the other. This causes the students to put more effort into rehearsal and performance.

The personnel are chosen, a robested plan is drawn up, a stage crew is selected, and the polishing begins. Each act is responsible for getting its own prope and costumes, for setting the stage, before the scone and clearing it

afterwards. The lights, curtains, and prompting are handled by a crew selected from the classes. Thirty or forty students participate in this hour program.

The acts are composed of the best pentonines, monologues, and character sketches. They are original.

Also at Whitefish Bay High School the senior speech classes present assemblies. These assemblies consist of assembly and programs for special days.

The outside talent programs are introduced by a carefully selected and a thoroughly prepared student chairman. 5

The assembly at Whitefish Bay High School certainly seems to afford an outlet for many of the speech activities.

Their plan appears to meet criterion four in that their everall program is planned by students and supervised by a wail qualified teacher. From the information about the preparation of the speech class assemblies, it appears that the programs become the responsibility of each class or club putting on the program. We do not know then if each program is well organised and well produced or not. The example of the speech program would lead us to believe that the programs are well organized and well produced. The author wishes to point out that it could be expected that the speech department programs would ordinarily be well planned, for it is generally part of the training of the discipling. What about the programs coming from the other

Leis Ramstack, "Assembly via the Classroom," Clearing House, 30(December, 1955), 202-204.

eress of the school? Who sees that they are well organised, well planned by students, well rehearsed, and well produced? There is an obvious lack here in connection with criteria four and five.

This progress partially neets criterion one. It appears to have a scheduled time for production. The rehearing seems to be done during the class period of the class producing the progress. When are the club progress which have no time in the scheduled day planned and reheareed?

Criterion seven, eight and nine seem to be met in that
the program emphasizes student participation and control,
the programs appear to enrich the educational experience of
the students, and they are interesting to a degree. It should
be pointed out, however, that one hour of pantomines, monologues, and character sketches could be boring.

The committee works for variety in the programs by utilizing all the school's activities. They cannot be sure there will be variety of an absence of monotony on each program. The creativity and resourcefulness of the students are utilized to an extent. The author feels that this is not fully done. She feels that this could be done by giving trained students an opportunity and time to help create the program ideas, to rehearse the program and train the participants if needed. Criterion six, therefore, has not been fully realized.

This school has developed a good program up to a point.

The speech department programs, it seems, are good but what about the programs scheduled by clubs and other areas of the school? Who will make sure they are equally good?

The reader has probably noticed the obvious emission of any discussion of criterion eleven. This was done because none of the programs evaluated thus far meet this criterion to any degree.

Extensive research has revealed only one program that meets any part of criterion eleven. It is the program plan at Shorewood High School in Shorewood, Wisconsin.

In 1946 they had an elective senior speech-English class who were responsible for the assembly programs. The class was composed of twenty-five of the leaders from different areas of the school. At the beginning of each senseter a chalman, two secretaries, and a librarian are cleated.

Each spring a tentative assembly schedule was prepared for the ensuing year. Any school activity is given a chance to present a program. An attempt is made to keep programs of a similar type well scattered.

The members of this class publicized the programs, kept a record of the materials used on each programs and kept a record on the evaluations of each program.

During one year this class wrote and directed fourteen complete scripts, furnished continuity or narrative for eleven scripts to act as a unifying factor for the separate numbers in a program, and made up three advertising stunts. This class is not responsible for the staging. The production class is the "stage crew" for the assemblies.

The numbers of the class learned much from this experience and the assembly programs were probably better than they would have been without the class.

The existence of the class, however, did not insure that the program met all the criteria. The program partially meets criterion one, for it has a regularly scheduled time for planning and producing. The author was unable to determine from the available information when and where the rehearing for the programs was done or if the rehearing was supervised by the class. Decause no mention was made of how this was done, the author feels that the program lacks something here. Thus we cannot be sure that the program fully meets criterion one and four.

It is not clear whether the programs are well organized and well produced. No definite mention was made in the available information as to how much help in organizing and

Lena F. Chariton, "Shorewood's Assembly Training Class," The Journal of Education, 129(February, 1946), 48-50.

producing the class gave to other areas or clubs who were scheduled to give a program. The author would say, then, the program does not fully meet criterion five.

In regard to the criterion six, the author believes that the program falls down here in emphasizing creativity and resourcefulness by limiting the class to seniors. The author feels the class should be open to all three grade levels in the high school. This is necessary in order to capitalise on talent and training. The athletic and other departments do it. Why shouldn't it be done in something as vital and important to the school as the assembly program?

To be certain the activities of the class enriches the educational experiences of the class members, but it is not known whether the individual programs enrich the educational experience and stimulate the imagination of the audience.

Parts of criterion eleven have not been met by the Shorewood assembly training class. Although it is a regularly scheduled class whose function is to produce assembly programs, no mention was made in the information as to whether the members of the class train those who need training for a part on the assembly program. The author feels this training is an important part of the class program. The members of the class are trained by the well qualified teacher. They in turn train the participants of the programs,

for it is impossible for the teacher to train all the participants. This training and close supervision is necessary to insure educational, inspirational, and interesting assembly programs.

It is the writer's opinion that criterion eleven part two was violated when the Shorewood class wrote fourteen complete scripts and partially wrote eleven others. It appears to the author that the program is not utilizing program material and participants from all areas of the school. The creative ability of the other students in the school is not being used enough.

The purpose of criterion eleven is to insure that the ideas and talents of all the students in the school can be properly utilized and channeled into a well planned, well organized, well produced, and stable assembly program.

As has been already pointed out none of the programs evaluated has met all the criteria. If none of these programs meet the criteria, what kind of program will? This the author hopes to answer in the next chapter.

#### Chapter V

#### AN EVALUATION OF THE PROPOSED PLAN

Let's examine in detail a program for integrating speech activities with assembly programs. What are the advantages of such a program? What are its limitations? Here is the plan under examination.

The plan is one in which the assembly program is operated by an academic class. The members of this class have the responsibility of the assembly programs. This assembly class may be in one of several departments of the school; however, in the plan under examination the class is placed in the speech department. In this study the class has been called an assembly class. The class may be called by various other names which might give more status to the class.

The members of this class may come from all grade levels in a three-year high school. A prerequisite for this class may be one sememster of speech. The class should carry graduation credit. It may be possible to take the class for two years and receive credit each time. The class should be taught by a well-qualified speech teacher.

The responsibilities of the class members are many. Their major responsibility is to plan and produce the assembly programs for the entire year. The class may be organised this way.

At the beginning of each semester a chairman, a publicity secretary, an executive secretary, and a librarian are elected or appointed by the teacher.

The student chairman leads the class discussions.

The teacher, of course, is active in the discussion. The discussions may concern many things from the types of assembly programs that should be given to what is the best or proper way for the president of the Girl's League to introduce the visiting speaker for the Girl's League assembly program.

The student chairman does more than lead the class and discussions. His big job is to coordinate the class and see that all assignments are done. The chairman must realise the important parts of each assembly and concentrate on them. He must be able to have several committees working on different projects and yet keep the class a unit. He, under the supervision of the class teacher, assigns committees to each assembly, and it is his responsibility to see that their work progresses satisfactorily. The secret of success for such a class as this is the use of the committee system.

The publicity secretary writes all the notices which inform the school, through publication and otherwise, of approaching assemblies. These notices may serve definite purposes. They may give facts concerning who is presenting the assembly, the nature of the assembly, and the anticipated audience reaction. The publicity secretary may, also, give out forms on which the students evaluate the assembly program.

The executave secretary keeps the assembly calendar. Ence the calendar has been set up for the year, she notifies each department or club at least two weeks in advance of their approaching assembly program. She also keeps the record of the class committees and who is working on what project. As the specific acts or parts of each program are planned by the group giving the program and a representative from the assembly class, the executive secretary keeps a record of these acts and the participants. This information she should have at least five days before the scheduled assembly. From these records, this secretary makes a list of those to be excused from classes to take part in a rehearsal for an assembly. She sees that these lists are received by the classroom teachers from whose class students are to be excused. The executive secretary or the publicity secretary may write any letters which are necessary to the production of an assembly.

The class librarian is in charge of the assembly files. In these files are kept the evaluations of each assembly given. It, also, contains complete records of each production or assembly. A detailed account of each act, the participants, and reference material such as poems, songs, is kept by the librarian. Also in this file may be kept available source material for assembly programs including community sources, commercial sources such as Bell Telephone, and agencies from which programs may be hired.

The class may operate in this manner. Early in the fall the principal or director of student activities meets with the assembly class and instructor to go over the tentative assembly schedule that was set up in the spring. There may be thirty-seven assemblies, one for each school week. The assembly should be acheduled for a certain period and day each week. Each school activity is given a chance to present one program such ass drama club, music department, science department, athletic department, math department. The types of assemblies are considered and an attempt is made to keep programs of a similar nature well scattered. For example, the speech department may present its oral interpretation program in November; the drama department may present its one-act play in February; and the advanced drama class may pre-view its spring play in May.

The class and the principal acting as a committee may broadly outline the purposes of the assemblies for example, to educate, to inspire, and to entertain. With these purposes they may assign an overall purpose to each assembly. For example, the athletic department may be expected to give a program with the overall purpose of entertaining. The music department may be expected to more easily give an assembly which would be inspiring. This is not to say that each purpose would not be found in every program or that each program would not have a specific purpose.

The committee may leave an open date occasionally for any emergency or any out of school assembly they may deem valuable to the students. It will be the responsibility of the assembly class to see that these open dates are filled at least two weeks in advance. This may give an opportunity for the class members to write, cast, and produce a program entirely from their class, an opportunity for a commercial assembly such as, a science demonstration from General Motors Corporation, a paid assembly from an agency such as, a person who is an authority speaking on rockets, or it may be an opportunity to present the personnel director of the local corporation who may give ideas on how to be a success in business. These programs from outside the school should as far as possible be previewed by a committee from the assembly class. Care should be taken to make sure these programs meet the standards and purposes of assembly programs.

when the assembly schedule has been made and each department or club has been notified of the date and purpose of its assembly, the class begins work on individual programs.

Class discussions directed by the chairman are held.

From these discussions the procedure for the class is established. The class procedure may be that on Monday the class will through discussion evaluate the last assembly.

These discussions should be held the class meeting immediately following the presentation of the assembly. Tuesday each

committee may work on whatever assembly to which it is assigned. Vednesday may be used for the members of various committees to present their problems and ask for help and suggestions from the class and instructor. Thursday may also be used for the committees to work on individual programs. Friday may be used for more questions and help on individual programs, lectures by the teacher on various phases of speech, such as, voice and diction, getting good speech materials, delivery and presentation, reports or speeches from students on such topics as, the importance of good programs, the sources for good programs, the types of good programs, variety in good programs, how oral interpretation can be worked into a good program. This day or one day should be devoted to development of the discipline of speech.

when the procedure for the class has been set up, the chairman end teacher assign committees for individual assembly programs. The committees may be established on a voluntary basis. Committees may consist of from three to five students. Committees for all programs should not be assigned at this time. As far as possible each member of the class should be permitted to work with every other member of the class.

A committee has been assigned to plan and produce with the music and speech departments the Easter program. The committee may proceed this way. A committee chairman may be chosen. His duties are numerous; some are: check and mark scripts for stage direction, set reheared schedule, inspire the cost with the right attitude toward their assembly, and check the technicalities with the stage ranager, who is a member of his committee.

The committee should then meet during the class period with a committee and faculty representative from the music and speech departments to develop an idea for the ascembly. They may also plan the thome, the purpose, and the acts of the program. When this has tentatively been done, the committee from the assembly class takes these ideas to the dicussion held in class. Here they may get instruction, approval, and suggestions from the teacher and class members. The class committee then meets again with the department consittee. Together they work out the details of the program. The progrem is then cast by the committees with the approval of the speech and music department faculty advisors. A rehearsal schedule is set up, and the program goes into rehearsal. The executive secretary will notify the teachers of those who are to be excused from class. Rehearsal for the program is held during the hour in which the assembly class meets.

The committees may have developed the following programs
The stage may be set with either pictures or real busny

rabbits, Laster hats, or some Easter representation from another country. There may also be a picture of the cross or Jesus.

Themes Easter

Title: "The Faces of Easter"

Purposes To inspire

Length: Approximately fifty minutes

Cpening: Sand playing "Feter Cottontail" or some other . light Easter song.

Introductions A student chairman explains that the program will be in three parts: the gay look of Easter, the foreign look of Easter, and the serious look of Easter. The program will begin with the gay look.

Shits Should bring out the way children see Easter, the Laster bunny, and the idea that Easter is the time for new clothes. The skit should be brief, humorous, and interesting. A humorous reading may be used here instead of the skit.

Musics Chair sings: "Easter Parade"

Chairman: Introduces the foreign look of Easter. These speeches should be written as continuity.

Speech: May be a five minute speech telling the way people in other lands spend Easter. It may include a story of how one person in a foreign country spent Easter.

Masic: Hay be an instrumental solo, duet, etc., The music should relate to the preceding part and lead into the next section.

Chairman: Introduces the serious look of Easter.

Speech: The Easter story including Scripture is given by a good speaker.

The Holy City" sung by a soloist or an insica

arrangement with the choir and soloist.

"The Crucifixion" by Lew Wellace, a cutting Readings

from Ben Hur interpreted by a good reader.

The choir may sing "Awe Marie" or "Fairest Lord Jesus." This closes the program. Pinaic:

It will be the duty of the assembly class committee to be sure that each person on the program is well prepared. During the rehearsal time the members of the cosmittee help the speakers write and practice their speaches. The speeches and readings may be practiced before the assembly class. This will give an opportunity for suggestions from the class and the speech teacher. The music will be prepared by the music department. The committee will check to be sure that all are in preparation and ready on the assembly date. The stage crew from this assembly committee will be sure the stage propa, lighting, etc. are ready. They will poerate these things during the progress.

At the first assembly class meeting following the program. the class will evaluate it. During the discussion the effective and ineffective features should be pointed out. Also, there should be a discussion as to why this was good or why it was bed, what would improve the program, and how it could be done.

There may be several committees working on programs at the same time. Heny types of programs may be developed. For example, one program which may have been developed in the above described way is "The Voice of Science." The general science class may present as a radio program having a studio audience a group of varied topics on phases of science surveyed in the year's work. "Dr. Einmets," a class member in tail coat and long gray wig, first impresses the audience with his vast imoviedge of the starry heavens. Hext, doctors of community hold a panel discussion about local disease control. A third group at the microphone are recognized as a city council concerned about the water supply. A "man from the soo" then speaks on the habits of the fismings or the hippopotamus. Last, a group reads an original radio script on the life of Edison. The acts are interspersed with typical radio announcements. Typical signs and signals of a radio broadcast may be used.

Another program which may be used is a debate. The author has used a debate as a high school assembly program, and it proved successful. A student chairman was used to introduce the proposition and the speakers. Much interest was around. A change of opinion ballot may be used to find out if the debate was successful in getting opinions changed. Also, it creates interest.

This program may be given by the social studies department and supervised, of course, by the assembly class. It is a "Tribute to the Past" of the community. Such a subject offers a broad field for practice in the subject type of

speech. At the same time it affords an excellent chance for furthering good relations between school and community. The use of the community as a inhoratory also permits practice in the interview and in conversation. The program may honor leaders and accomplishments in every field, e. g. politics, industry, art. Pageantry may be worked into the program. The highlight of the program may be a short talk by an old settler of the community.

The author has used, also, the following program which proved to be interesting and successful: A student chairman, from the speech class, introduced the program. The program was composed of orations, interpretative orations, humorous interpretation, and dramatic interpretation. These students were the local forensic contest winners. There were eight readers. The readings were arranged on the program so as to keep interest. The length of the program was approximately fifty minutes.

The possibilities for good programs, programs integrated with speech activities are unlimited under the guidance and supervision of an essembly class.

An examination of the plan must also include an evaluation of the plan in the light of the criteria set up in Chapter III. The evaluation will include both advantages and limitations of the program.

## Criterian Cne

A good assembly has a regularly scheduled time for planning, rehearing, and producing.

The proposed plan meets this criterion for the assembly is regularly scheduled by the principal; it is regularly planned and rehearsed by the assembly class. It should be pointed out here that there are limitations to the program in connection with rehearsing. It may be necessary to rehearse the assembly programs at some other time than during the assembly class period. The problem of rehearsel has not been entirely solved.

## Criterion Two

A good assembly has adequate equipment and facilities.

It is impossible to determine if this program meets this criterion. Here is another limitation of the program.

# Criterion Three

A good assembly is suitable for the length of time, the place, the facilities, the purposes, and the occasion.

Again it is not certain if the program meets all this criterion. The program described above has a purpose and is suitable for the occasion. The length of time, the place, and the facilities can only be determined by individual achools.

## Criterion Four

A good essembly is planned and rehearsed with a definite purpose in mind by students who are supervised by a well qualified and cooperative teacher.

The program examined meets this criterion. It is planned and rehearsed by students with a definite purpose in mind. The speech teacher who is, all things being equal, well-qualified supervised the program. It is not known if the teacher was cooperative. It should be pointed out here that the examined program does not solve all the assembly problems, but it is believed that given the same opportunities and the same caliber of students that the problems will be fewer and the chances of success will be greater with this program in operation.

## Criterion Five

A good assembly is well organized and well produced thus reducing disciplinary action to a minimum.

The specific program examined here appeared to be well organized, and it is believed the chances of a well produced program are created with this type of essembly practice.

# Criterion Six

A good essembly places emphasis on creativity, resourcefulness, and variety by utilizing all the activities of the school.

one advantage of the program being examined is that the emphasis is placed on creativity and resourcefulness. The students are given time and opportunity to use their creative ability in developing the programs. There is a variety of programs and they are well spaced when the programs for the year are planned at the beginning of the school year. The resourcefulness of the entire school is utilised to a greater extent in this program.

There are, however, limitations in this area. No mention of "pop" assemblies is made in this program. Hany schools hold "pep" assemblies every friday. Are these "pep" assemblies to be considered in the regular assembly schedule or are they held some other time? It may be possible to hold "pep" assemblies during nutrition period, during lunch period, or after school. This problem has not been solved by the above program.

## Criterion Seven

A good assembly emphasizes student participation and control.

The above program appears to meet this criterion. The students participate in the programs and control the production.

# Criterion Eight

A good assembly is interesting and answers a definite need of the students.

The program appears to be interesting. The real test of an interesting program, however, is how the audience reacts to it. It appears, also, to consider the moeds of students. Here again is a limitation of the program, for it is not known whether the programs will be interesting and meet the needs of the students. It appears that the chances of this criterion being met are greater with this program.

# Criterion Mine

A good assembly enriches the educational experience of the participants and sudience and stimulates their imagination.

This program enriches the educational experience of the participants in that the students have a great part in developing the program. It is felt that imagination would be stimulated in the preparation and production of the program. Whether the program is educationally beneficial to the audience or stimulates their imagination, depends upon each program. The one being examined appears to do this.

## Criterion Ten

A good assembly engenders wholesome and mutually beneficial community-school relationships.

This program, all things being equal, should do this. Good public relations would be build as the students study the community for possible program material. Contacts in the community will be made for possible speakers or to borrow stage props. These contacts should engender good relationships, for the students in the assembly class are in training to meet the public.

# Criterion Eleven

A good assembly must be the outgrowth of a regularly constituted ecademic class, the members of which plan and operate the assembly plan.  The members of this class, supervised by a qualified teacher, plan, organize, and manage the program in addition to training those who will participate in the programs.

 The programs, although supervised by an academic class, are all-school assemblies with the participants and program material coming from

all areas of the school.

The program being examined met this criterion. It is felt that the absence of criterion eleven is the crux of the problem of assembly programs. Good assembly programs may be given using other plans, but it is believed that the proportion of good, worthwhile assemblies will be greater if this program is used.

The program evaluated here is the outgrowth of a regularly constituted academic class. The students of the
class plan, organize, rehearse, and produce with students
from various departments and clubs the assembly programs.
Opportunity is given for all areas of the school to be
represented on a program. Many students are given the
opportunity to participate. These activities are supervised by a qualified teacher.

in this plan the assembly class has been placed in the speech department. The class could be located in one of several departments. For example, it could be a part of the music department. It, also, could be a part of the English or social studies departments. However, after

exemining the possibilities the speech department seems to be the logical place for it.

one reason it is feit that this class should be in the speech department the students are in training to meet the public. They are trained to prepare and produce programs.

Speech activities need an outlet. Given an outlet such as the assembly program these activities will be much more meaningful.

This program in the speech department should give the student body a better quality assembly program.

This plan is not offered as a totality. The proposed scadenic assembly class located in the speech department is offered as one way to give purpose, importance, dignity, and stability to the assembly programs, thereby, giving rise to the third stage in the development of secondary school assembly programs.

# Chapter VI

The problem of integrating the public secondary school assembly program and speech activities was undertaken not in the sense of a final pronouncement on assembly program plans but rather to create an assembly plan inductively built on current educational philosophy for high school assembly programs.

That the high school essembly program is an important, educational part of the school today was established. Authorities agree that this activity in which all students and faculty are brought together has a vital role to play in the high school.

problems. Reports from surveys of assembly practices reveal the greatest problems to be in the areas of: (1) the time for the assembly; (2) the place and equipment for the assembly; (3) the planning and organizing of the assembly; (4) the arranging of suitable programs; (5) the interest and co-operation of teachers who are trained for the job.

It was seen that the development of the assembly program has been rather slow, and there seem to have been

two very distinct periods in the history of this activity.

In its early development the programs were entirely religious.

They were presided over by the principal and the students

in this "chapel service" were to be seen and not heard.

Cradually, and with some overlapping, the second period emerged. In this period the students were given an opportunity to participate in the programs. The concept of student participation has evolved and unfolded until today, in many cases, the students not only participate but also plan, organice, and produce the assembly programs.

This has been a materal development. As philosophy of education has changed so has the purpose for the essembly programs. As the emphasis in education continues to shift, the author believes that the third stage in the development of the assembly program is inevitable. This step will be to make the assembly program the responsibility of the academic curricular class.

The assembly program should be planned by someone who is interested in making it an educational force. If this program is an important and vital as research reveals, then it should be given a place in the curriculum. Curriculum / standing would give purpose, dignity, and stability to the assembly program.

This proposed scademic assembly planning class could be placed in many departments of the high school. It could be

part of the music department, the English department, or the social studies department. After careful consideration of the possibilities, it is felt that the speech department is the most logical place for the class.

It is in the speech department that students are being trained to meet the public. It has been established that there is a need for practice in oral English in the high schools today. The consensus is that given a place to demonstrate what has been learned will render the activities of any class more meaningful. Speech activities need the life-like situations of the assembly program. It is believed that assembly programs planned and produced by an academic class in the speech department will give to the student body superior programs.

in addition to the investigation of the history and development of assembly programs, a point of view concerning the philosophy of education and assembly programs was developed as a basis for establishing the objectives of assembly programs today and setting up criteria by which assembly programs could be availated. Following the investigation of philosophical assumptions which stand behind assembly programs, eleven criteria wars recommended as a basis for setting up and evaluating an assembly program plan.

In view of the criteria, essembly programs from six high schools were chosen on the basis of location, type, and their apparent attempt to integrate speech activities to some entent with assembly programs. These schools are:

(1) Peacock Military Academy, San Antonio, Texas; (2)

Wakefield High School, Arlington, Virginia; (3) Community

Migh School, Granite City, Illinois; (4) Boulder High School,

Boulder, Colorado; (5) Whitelish Bay High School, Milwaukee,

Wisconsing and (6) Shorewood High School, Shorewood, Wisconsin.

It was found that although these schools seem to have worked out a comparatively successful program none of them
successfully meet all the criteria. All the criteria were
met in varying degrees in their efforts to integrate speech
activities with the assembly program.

Finally, a program was developed which the author believes will successfully integrate assembly programs and speech activities. The details and operation of the program were outlined. It was established that this program meets all the criteria.

This plan is not offered as a stopia. What the author has done here has been in the reals of hypothesis and out of limited background and experience. The effectiveness of the plan has yet to be proved.

There are other approaches to this problem; there should be explored. Assembly progress problems are a part of our educational system, and teachers and educators with a vision should try to find a solution.

It is boped that this study will point the way for other studies that will perhaps establish the educational value of the assembly program in the public secondary schools of today.

#### BIBLICERAPHY

#### BOOKS

- Adler, Hortimer J., and Hayer, Milton, The Revolution in Education. Chicagos The University of Chicago Press, 1958.
- Bent, Rydyard Ke, and Kremenberg, Henry H., <u>Principles of Secondary Education</u>, Hew Yorks McGreme-Hill Book Company, 1959
- Bossing, Nelson L., Principles of Secondary Education.
  New Yorks Prentice-Hall, Inc., 1949.
- Perspective. New York: Dryden Press, 1955.
- Doughton, Isaac, Hodern Public Educations Its Philosophy and Background, New Yorks Appleton-Century-Crofts, 1935.
- Douglass, Hari R., Modern Administration of Secondary Schools. Chicagos Gin and Company, 1951.
  - Flaum, Laurence S., The Activity School. New York: Harper and Brothers, 1953.
- French, William M., American Secondary Education. New York: The MacHillan Company, 1951.
- French, Will, et. al., Behavioral Coals of General Education in High Schools, New Yorks Russell Sage Foundation, 1957.
- Prench, Villian, Great Issues in Education. Chicagos The Great Books Foundation, 1956.

- Fretweil, E. R., Extra-Curricular Activities in Secondary Schools. New York: Houghton Fifflin Company, 1931.
- Committee. Cambridge, 1915.
- Hand, Harold C., Principles of Public Secondary Education.
  Hew Yorks Harcourt, Brace and Company, 1958
- Henderson, Stella Van Petten, <u>Introduction to Philosophy</u>
  of Education. Chicagos University of Chicago, 1947.
- Rilpetrick, William H., Philosophy of Education. New Yorks The NacMillen Company, 1951.
- Kilser, Louis R., Stephenson, Harold H., and Nordberg, H., Orville, Allied Activities in the Secondary School, New York: Harper and Brothers, 1956.
- Lanks, T. A., and Silvey, Herbert M., Masters Theses in Education. (Annually since 1951).
- McRoss, Herry, Assembly and Auditorium Activities, New Yorks The MacMillan Company, 1930.
- Hiller, Franklin, Hoyer, James H., and Patrick, Robert B., <u>Plenning Student Activities</u>. New Jersey: Prentice-Hall, Inc., 1956.
- Planning Better Programs. Leadership Pamphlet no. 2.

  Adult Association of the United States of America.

  Chicago, 1956.
- Sayers, Ephraim V., and Madden, Ward, Education end the <u>Democratic faiths An Introduction to Philosophy of</u> <u>Education</u>. Hew Yorks Appleton-Century-Crofts, 1959.
- A School Assemblics Handbook. Department of Speech, Western Hichigan University, Kalamaso, 1957.

- Thompson, Rellie Lette, <u>Vitalizai Assemblica: 200 Programs</u>
  <u>for All Geossions</u>. New Yorks E. P. Button, 1952.
- Vagner, H. Cheaning, Appenbly Progress. New Yorks A. S. Daraes and Company, 1950.

#### PETHICITALS

- Amer, J. Jeffery, "Doctoral Dissertations in Speech and Drawn." Speech Honographs. (Annually since 1951).
- Dischurn, Mary, "The Speech Teacher and the High School Assembly Program," The Counterty Journal of Speech. 35(February, 1927).
- Chariton, Lone F. "Shorewood's Assembly Training Class."

  The Journal of Education. 129(February, 1946).
- Does Clyde W. "Abstracts of Theses in Speech and Drama."

  Speech Noncorpobs. (Annually since 1950).
- Evans, C. E., "what to no with the High School Assumbly." School Peview. 21(April, 1925).
- Pretiell, S. M., \*Estra-Curricular Activities of Secondary Schools.\* Teachers College Second. 25(Nay. 1926).
- Association of Secondary School Principles. Sixth Yearbook, 1932.
- Forvey, C. C., et. al., "The Assembly Program in the Secondary School." The Bulletin of the National Association of Secondary School Principals. School Resembles, 1946).
- Field of Speech. Speech Honographs. (Annually since 1905).

- Mnower, Fronklin H., "A Philosophy of Speech for the Secondary School." Speech Teacher. 1(Harch, 1952).
- The Bulletin of the National Association of Secondary School Principals, S2(January, 1913).
- Morgan, John D., "Auditorium Is Teaching Unit." <u>Metion's</u>
  Schools. 25(Movamber, 1940).
- Cliver, Robert T., "The Place of Speech in Education."

  School and Society. 86(January 4, 1958).
- Petton, Robert L., "Getting All Students Into the Act."
  School Frecutive. 76(November, 1986).
- Paul, Margaret D., "The Teaching of Speech thru the Aud-Itorium Mathod in Gary." The quarterly Journal of Speech. (April, 1927).
- Ramstach, Lois, "Assembly via the Classroom." <u>Clerring</u>
  <u>Prime</u>. 30(December, 1955).
- Richardson, Lawson H., "An Evaluation of Integrating the Assembly with Curricular Activities." The Bulletin of the National Association of Secondary School Frincipals. Si(November, 1757).
- Rountsly, Fredie, "Practical Aspects of the High School Assertly." The Cuarterly Journal of Speech. 21(February, 1935).
- Skinner, Theodore, "Suggestions for the High School Assembly."

  The Questerly Journal of Speech. 33(December, 1947).
- Thomas, Ruth, "Auditorium Programs." The Quarterly Journal of Speech. \$2(February, 1936).

- Wenrick, Thomas, "Auditorium Class: A Problem-Solver at McKinley Junior High." <u>Clearing House</u>. 13(December, 1938).
- Whitlock, James W., "Organization and Administration of the School Assembly." School Activities. 28(May, 1957).
- School Activities. 28(April, 1957).
- Wilson, Harold H., "Programs with a Learning Function."

  School Executive. 76(November, 1956).

#### UNPUBLISHED MATERIAL

- Pinley, Georgia I., "The Value of Trained Anditorium Class Programs as an Aid for Improved Assembly Programs." Unpublished Master's thesis. Marquette University, Milwaukee, 1937.
- Hardy, Myrtle, "The Place of Speech Arts in Anditorium Activities of the Platoon School." Unpublished Master's thesis. University of Southern California, Los Angeles, 1933.
- Mudeen, Dexie, "Development of a Proposed Auditorium Program."
  Unpublished Master's thesis. Southwest Texas State
  Teachers College, San Marcas, 1955.
- King, Mary B., "A Study Concerned with the Integration of Speech Training with the Social Studies Program to Influence Social Adjustment." Unpublished Master's thesis. Boston University, Boston, 1954.
- Sellers, Geneview T., "Speech as an Educational Factor in the Secondary School Assembly." Unpublished Master's thesis. University of Southern California, Los Angeles, 1934.

